Author's response to reviews

Title: Antinociceptive tolerance effects of NSAIDs microinjected into dorsal hippocampus

Authors:

Gulnazi Gurtskaia (gurtskaiaguliko@yahoo.com)
Nana Tsiklauri (nanatsiklauri1812@gmail.com)
Ivliane Nozadze (nozadzelia@yahoo.com)
Marina Nebieridze (mnebo@mail.ru)
Merab G Tsagareli (tsagareli@biphysiol.ge)

Version: 2 Date: 13 November 2013

Author's response to reviews:

In the title page I provided email addresses of authors:
GG: gurtskaiaguliko@yahoo.com
NT: nanatsiklauri1812@gmail.com
IN: nozadzelia@yahoo.com
MN: mnebo@mail.ru
MGT: tsagareli@biphysiol.ge;
I have formatted the abstract according to your guidelines for authors.
I have placed the Conclusion after the Discussion.
I have changed the Disclosure to Competing Interests.
I have added: Authors’ contributions
GG, NT, and IN equally contributed to data collection. MN carried out histological control. MGT designed conception, finally analyzed the data and drafted the manuscript. All authors read, contributed to and approved the final manuscript.